

GOPEN ACCESS

Citation: Wibaek R, Vistisen D, Girma T, Admassu B, Abera M, Abdissa A, et al. (2019) Associations of fat mass and fat-free mass accretion in infancy with body composition and cardiometabolic risk markers at 5 years: The Ethiopian iABC birth cohort study. PLoS Med 16(8): e1002888. https://doi.org/10.1371/journal.pmed.1002888

Academic Editor: Kathleen Rasmussen, Cornell University, UNITED STATES

Received: March 14, 2019

Accepted: July 16, 2019

Published: August 20, 2019

Copyright: © 2019 Wibaek et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The iABC Study is part of the Jimma University Clinical and Nutrition (JUCAN) Research Partnership, Jimma, Ethiopia (https://www.ju.edu.et/jucan/). The iABC study data can be made available upon request directed to the JUCAN Steering Committee currently headed by Dr. Melkamu Berhane (melkamuberhane@yahoo. com). Data cannot be made available in a public repository due to legal and ethical restraints. The informed consent for the iABC study was collected RESEARCH ARTICLE

Associations of fat mass and fat-free mass accretion in infancy with body composition and cardiometabolic risk markers at 5 years: The Ethiopian iABC birth cohort study

Rasmus Wibaek^{1,2*}, Dorte Vistisen², Tsinuel Girma^{3,4}, Bitiya Admassu^{1,4,5}, Mubarek Abera^{1,4,6}, Alemseged Abdissa^{4,7}, Marit E. Jørgensen^{2,8}, Pernille Kæstel¹, Kim F. Michaelsen¹, Henrik Friis¹, Jonathan C. K. Wells⁹, Gregers S. Andersen²

Department of Nutrition, Exercise and Sports, University of Copenhagen, Copenhagen, Denmark,
Clinical Epidemiology, Steno Diabetes Center Copenhagen, Gentofte, Denmark,
Department of
Paediatrics and Child Health, Jimma University, Jimma, Ethiopia,
Jimma University Clinical and Nutrition
Research Partnership (JUCAN), Jimma University, Jimma, Ethiopia,
Department of Population and Family
Health, Jimma University, Jimma, Ethiopia,
Department of Population and Family
Health, Jimma University, Jimma, Ethiopia,
Department of Laboratory Sciences and Pathology, Jimma University, Jimma, Ethiopia,
National Institute of Public Health, Southern Denmark University, Copenhagen, Denmark,
Childhood Nutrition Research Centre, UCL Great Ormond Street Institute of Child Health, London, United Kingdom

* rwibaek@gmail.com

Abstract

Background

Accelerated growth in early childhood is an established risk factor for later obesity and cardiometabolic disease, but the relative importance of fat mass (FM) and fat-free mass (FFM) accretion is not well understood. We aimed to study how FM and FFM at birth and their accretion during infancy were associated with body composition and cardiometabolic risk markers at 5 years.

Methods and findings

Healthy children born at term were enrolled in the Infant Anthropometry and Body Composition (iABC) birth cohort between December 2008 and October 2012 at Jimma University Specialized Hospital in the city of Jimma, Ethiopia. FM and FFM were assessed using air displacement plethysmography a median of 6 times between birth and 6 months of age. In 507 children, we estimated individual FM and FFM at birth and their accretion over 0–3 and 3–6 months of age using linear-spline mixed-effects modelling. We analysed associations of FM and FFM at birth and their accretion in infancy with height, waist circumference, FM, FFM, and cardiometabolic risk markers at 5 years using multiple linear regression analysis. A total of 340 children were studied at the 5-year follow-up (mean age: 60.0 months; girls: 50.3%; mean wealth index: 45.5 out of 100; breastfeeding status at 4.5 to 6 months postpartum: 12.5% exclusive, 21.4% almost exclusive, 60.6% predominant, 5.5% partial/none). Higher FM accretion in infancy was associated with higher FM and waist circumference at 5 years. For instance, 100-g/month higher FM accretion in the periods 0–3 and 3–6 months was associated with 339 g (95% CI: 243–435 g, p < 0.001) and 367 g (95% CI: 250–484 g, p

in 2008-2011 and public sharing of any individual level data was not part of the consent at that time.

Funding: This research was supported by The Danish Council for Strategic Research, Program Commission on Food and Health, by Danida through the Consultative Research Committee for Development Research (Recipients: GSA; Grant number: 104.Dan.8-1207; url: https://ufm.dk/en/ research-and-innovation/councils-andcommissions/former-councils-and-commissions/ the-danish-council-for-strategic-research), Danish Diabetes Academy (Recipient: RW; url: https:// www.danishdiabetesacademy.dk/), and Innovation Fund Denmark (Recipient: RW; Grant number: 501600082B; url: https://innovationsfonden.dk/en). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: ADP, air displacement plethysmograph; BC, body composition; FFM, fatfree mass; FM, fat mass; HDL, high-density lipoprotein; iABC, Infant Anthropometry and Body Composition; IWI, International Wealth Index; LDL, low-density lipoprotein; LSME, linear-spline mixedeffects. < 0.001) greater FM at 5 years, respectively. Higher FM at birth and FM accretion from 0 to 3 months were associated with higher FFM and cholesterol concentrations at 5 years. Associations for cholesterol were strongest for low-density lipoprotein (LDL)–cholesterol, and remained significant after adjusting for current FM. A 100-g higher FM at birth and 100-g/ month higher FM accretion from 0 to 3 months were associated with 0.16 mmol/l (95% CI: 0.05-0.26 mmol/l, p = 0.005) and 0.06 mmol/l (95% CI: 0.01-0.12 mmol/l, p = 0.016) higher LDL-cholesterol at 5 years, respectively. Higher FFM at birth and FFM accretion in infancy were associated with higher FM, FFM, waist circumference, and height at 5 years. For instance, 100-g/month higher FFM accretion in the periods 0–3 and 3–6 months was associated with 1,002 g (95% CI: 815–1,189 g, p < 0.001) and 624 g (95% CI: 419–829 g, p < 0.001) greater FFM at 5 years, respectively. We found no associations of FM and FFM growth with any of the other studied cardiometabolic markers including glucose, HbA1c, insulin, C-peptide, HOMA-IR, triglycerides, and blood pressure. Non-attendance at the 5-year follow-up visit was the main limitation of this study, which may have introduced selection bias and limited the power of the regression analyses.

Conclusions

FM accretion in early life was positively associated with markers of adiposity and lipid metabolism, but not with blood pressure and cardiometabolic markers related to glucose homeostasis. FFM accretion was primarily related to linear growth and FFM at 5 years.

Author summary

Why was this study done?

- Suboptimal growth in fetal life and early childhood are important risk factors for later obesity, diabetes, and heart disease.
- Progression towards disease may start early in life, demonstrated by studies linking early growth with changes in fat mass, lean mass, and cardiometabolic risk markers in children as young as 4 years of age.
- In populations from low-income countries, currently experiencing a rapidly growing chronic disease burden, knowledge about these relationships is lacking.

What did the researchers do and find?

- We followed a cohort of healthy preschool children from the city of Jimma, Ethiopia, for 5 years starting from birth.
- Using advanced statistical growth modelling and accurate assessment of body composition, we estimated individual fat and lean mass accretion in the periods 0–3 and 3–6 months of age in 507 children.
- We conducted regression analyses to test for associations of fat and lean mass growth in early infancy with body composition and cardiometabolic risk markers measured at 5

years of age. These analyses showed that children who gained more fat mass in early infancy had greater fat mass and waist circumference and higher concentrations of risk markers related to cholesterol metabolism at 5 years. Children who gained more lean mass in early infancy were taller and had more lean mass at 5 years.

• Fat and lean mass growth in early infancy were not associated with blood pressure and cardiometabolic markers related to glucose homeostasis at 5 years.

What do these findings mean?

- This study showed that early infancy is an important developmental window for a child's later body composition.
- This contemporary cohort of preschool children from a sub-Saharan African urban setting had markedly higher fat mass than similar-aged children from the UK and India, which may be a result of excess fat mass accretion during the first 6 months of life.
- Variable patterns of fat and lean mass growth did not appear to be consistently associated with markers of cardiometabolic risk, other than markers of cholesterol metabolism at 5 years.

Introduction

Non-communicable diseases like type-2-diabetes and cardiovascular diseases are among the leading causes of death and disability worldwide [1,2]. There is mounting evidence that perturbations in fetal and early-life growth increase the risk of a wide range of metabolic disorders such as obesity, type 2 diabetes, and cardiovascular disease in adulthood [3–6]. Over the last 3 decades the research focus has shifted from the consequences of fetal growth restriction, indexed by low birth weight, to the harmful effects of childhood obesity and rapid early growth on a variety of later health outcomes including body size, body composition (BC), and the risk of cardiometabolic diseases. The relative importance of fetal versus postnatal growth has been debated [7,8], but it is likely that both aspects of early growth are involved in the risk development [9,10]. Most studies relating early growth to later health have been conducted in high-income populations. However, as many low- and middle-income countries are currently undergoing rapid nutritional transition and more than 80% of the global mortality burden of non-communicable diseases already occurs in low- and middle-income countries [11], it has become increasingly important to identify critical windows of growth associated with obesity and risk of cardiometabolic diseases in these populations.

Simple anthropometry, such as weight or length, is commonly used to study associations of early growth with later health outcomes. However, as the 2 main components of the growing child, fat mass (FM) and fat-free mass (FFM), have different metabolic properties [12], they most likely contribute differently to the programming of later cardiometabolic disease risk as well as the growth and development in early childhood.

In the Infant Anthropometry and Body Composition (iABC) cohort of more than 500 Ethiopian children with repeated measurements of FM and FFM from birth to 6 months of age, we have previously reported different proportions of FM and FFM among newborns with similar body weights [13], and identified very distinct patterns of FM and FFM growth through infancy [14]. Contrasting accretion patterns in the metabolically diverse tissues of FM and FFM are likely to play a key role in the associations between early growth patterns and later cardiometabolic disease risk. We therefore aimed to study the associations of FM and FFM at birth and their accretion during infancy with height, waist circumference, FM, FFM, and cardiometabolic risk markers at 5 years of age in Ethiopian children.

Methods

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 Text</u>). Data for the present study were collected following a prospectively written study protocol (<u>S2 Text</u>).

Study setting and participants

The iABC study is a prospective birth cohort study of the determinants and consequences of growth variability in early childhood [13,15]. The study was carried out at Jimma University Specialized Hospital in Jimma, Ethiopia (with a population size of 157,432 [16], and situated 350 km southwest of the capital Addis Ababa). Mother–child pairs meeting our eligibility criteria (residing in Jimma, gestational age at birth \geq 37 completed weeks of pregnancy, birth weight \geq 1,500 g, no congenital malformations) were enrolled between 17 December 2008 and 24 October 2012. Eligible mother–child pairs were examined within 48 hours of birth and were invited for a total of 12 scheduled visits between birth and 5 years of age. To estimate FM and FFM accretion in infancy, we used data on FM and FFM at birth and at 1.5, 2.5, 3.5, 4.5, and 6 months of age. To capture the dynamics of FM and FFM accretion in early infancy, we required a minimum of 3 assessments of FM and FFM between 0 and 6 months, including an assessment at birth, to be included in the BC growth modelling. The outcome data on BC and cardiometabolic markers were collected at the 5-year visit.

Data collection

Anthropometry and BC in infancy and early childhood. Standing height at 5 years was measured in duplicate to the nearest 0.1 cm (model 213 stadiometer, Seca, Hamburg, Germany). Weight, FM, and FFM from birth to 6 months were assessed with a PEA POD—an infant air displacement plethysmograph (ADP) designed to measure infants between birth and 6 months of age (COSMED, Rome, Italy). At 5 years, weight, FM, and FFM were assessed with a BOD POD—a child/adult ADP with a paediatric chair insert allowing accurate assessment in children above the age of 2 years (COSMED). These ADP instruments provide accurate, precise, feasible, and safe assessment of FM and FFM in infants and children [17-19]. The PEA POD has previously been validated in the iABC cohort against a 3-compartment model of BC incorporating measurement of total body water by stable isotopes [15]. A comprehensive overview of the theory and methods behind the PEA POD and BOD POD techniques is found elsewhere [20,21]. In short, an ADP relies on densitometry to distinguish the 2 body components FM and FFM. First, by measuring total body weight and volume, the total body density is derived. Subsequently, since the density of FM and FFM differs, the relationship between the total body density and the assumed densities of FM and FFM is used to attribute the body weight to either FM or FFM, using a 2-component model of BC and Archimedes' principle [22]. The density of FM was assumed to be constant at 0.9007 g/cm³, while age- and sex-specific densities of FFM were used [23]. The calculations were performed by the inbuilt computers of the PEA POD and the BOD POD, software versions 3.3.0 and 5.2.0, respectively. A complete BC assessment lasted 5-10 minutes, and the 2-minute volume measurement occurred in an enclosed transparent test chamber (PEA POD and BOD POD). In the PEA POD the nude infant was placed in supine position on a tray wearing a swim cap, and in

the BOD POD the child sat on a paediatric chair insert wearing a swim cap and tight fitted underpants.

Blood pressure at 5 years. After relaxing for a minimum of 5 minutes, systolic and diastolic blood pressure were measured in a sitting position using a blood pressure monitor with age-appropriate cuffs (Pressostabil model, Welch Allyn, Skaneateles Falls, NY, US). Measurements were done in duplicate, and the values averaged.

Other cardiometabolic markers at 5 years. After a minimum of 3 hours of fasting, 2 ml of venous blood was drawn from the antecubital fossa. We determined glucose concentrations from whole blood using the HemoCue Glucose 201 RT System (HemoCue, Ängelholm, Sweden). Glycosylated haemoglobin (HbA1c, mmol/mol) was determined from whole blood using a DCCT aligned Quo-Test A1c Analyzer (EKF Diagnostics, Cardiff, Wales). After clotting, the whole blood was centrifuged to isolate serum, divided into three 0.4-ml aliquots and frozen at -80 °C until analysed at the Ethiopian Public Health Institute, Addis Ababa, Ethiopia. Serum concentrations of total cholesterol, low-density lipoprotein (LDL)–cholesterol, high-density lipoprotein (HDL)–cholesterol, and triglycerides (all in mmol/l) were determined using the COBAS 6000, module c501, and insulin (μ U/ml) and C-peptide (ng/ml) concentrations were determined using the COBAS 6000, module c501, module e601 (Roche Diagnostics International, Rotkreuz, Switzerland). We calculated the homeostasis model assessment of insulin resistance index (HOMA-IR) as insulin × glucose/22.5 [24].

Covariates. Maternal postpartum height was measured in duplicate to the nearest 0.1 cm using a Seca 214 stadiometer (Seca, Hamburg, Germany). We used an average of the available measurements from birth to the 6-month visit. Data on birth order of the current child (parity), child's sex, gestational age at birth, maternal age, maternal educational level, and family socioeconomic status were collected through questionnaires at the birth visit. Gestational age at birth of the current child was assessed using the New Ballard Score test instrument [25]. Socioeconomic status of the family was estimated using the International Wealth Index (IWI). The IWI estimates the wealth status of families in low- and middle-income countries using 12 material well-being items, including 7 items on household assets, 2 items on access to public services, and 3 items on characteristics of the house [26]. The IWI has a range of 0 to 100 (highest wealth). Data on breastfeeding status were collected at the follow-up visits at 4.5 and 6 months after birth and divided into 4 categories: exclusive (no other foods given), almost exclusive (no other foods given except water), predominant (breast milk as primary food), and partial/none (breast milk not the primary food/not breastfeeding) [27]. We used the breastfeeding status at the 6-month visit, but if a child did not attend the 6-month visit we used the breastfeeding status from the visit at 4.5 months of age.

Ethics

The study was approved by the Ethical Review Committee of Jimma University (Reference RPGC/279/2013). Written, visual, and oral information about the study was presented in local language prior to obtaining written consent from a parent or caregiver. No risks were associated with the examinations, and a topical anaesthetic (EMLA cream) was used prior to collecting the 2 ml of venous blood sample. Medical conditions noticed by the research nurses were addressed according to local clinical guidelines.

Statistical methods

Descriptive data are presented as mean (standard deviation [SD]) or median (interquartile range) for continuous variables and count (percentage) for categorical variables. Differences between groups were tested by 1-way ANOVA F-test for continuous variables and Pearson's

chi-squared test of independence or Fisher's exact test of independence for categorical variables. Continuous variables with a right-skewed distribution were log-transformed (natural logarithm) prior to regression analyses. Estimates from these models were back-transformed and presented as percentwise change. A significance level of 5% was used. All analyses were carried out in R version 3.4.1 (R Foundation for Statistical Computing).

FM and FFM accretion in early life. Linear-spline mixed-effects (LSME) modelling was used to approximate the non-linear relationship of age with FM and FFM by deriving a number of child-specific and average summary measures of growth over discrete time intervals from 0 to 6 months of age [28,29]. LSME modelling differs from conventional mixed-effects modelling by combining 2 or more linear mixed-effects modelling functions at pre-specified ages (knot points). Thus, the estimated FM and FFM growth velocities are constant within a given time interval but allowed to differ between successive time intervals. Separate LSME models, specified with a knot point at 3 months of age, were fitted for FM and FFM. The LSME models for FM and FFM return 3 average and 3 child-specific growth parameters: estimated FM/FFM at birth and estimated FM/FFM growth velocity in the periods 0–3 and 3–6 months. The child-specific growth parameters are used as continuous exposure variables in the subsequent regression analyses of the BC and cardiometabolic outcomes. A detailed description of the modelling of FM and FFM growth velocity is provided in <u>S3 Text</u>.

Associations of FM and FFM accretion in early life with BC and cardiometabolic markers at 5 years. Associations of estimated FM and FFM at birth and their growth velocity over the periods 0-3 months and 3-6 months with cardiometabolic markers and BC at 5 years were analysed in separate multiple regression models (e.g., FM at 5 years regressed on estimated weight gain velocity from 0 to 3 months and adjusted for relevant covariates in separate models). Model 1 was adjusted for child's sex, birth order, gestational age at birth, child's exact age at the 5-year visit, maternal age at delivery, maternal postpartum height, maternal educational status, and IWI. Model 2 was additionally adjusted for FM at the 5-year visit. In the regression analyses of FM and waist circumference at 5 years as outcome, model 2 was adjusted for FFM at the 5-year visit instead of FM. We used a complete case approach, limiting the analyses to children with complete data on the included covariates. Covariates included in model 1 and 2 were identified a priori based on adjustment practices in similar studies and reported associations of the covariates with growth in early life and the outcomes studied [30-38]. To obtain comparable estimates across the different growth periods, exposure variables were standardised prior to the regression analyses. Thus, the estimates indicate the change in outcome per study population SD increase of the exposure variable (e.g., FM accretion from 0 to 3 months). We also present estimates for the change in outcome per 100-g higher birth FM and FFM and 100-g/month higher FM and FFM growth velocity over the periods 0-3 and 3-6 months of age. In additional analyses, we accounted for multiple testing using the Benjamini-Hochberg approach [39], with the number of tests set to 90 (15 outcomes and 3 age periods for the 2 exposures FM and FFM). Finally, as data on breastfeeding were only available on a smaller sub-sample, we ran sensitivity analyses on this sample where we adjusted model 1 and 2 for breastfeeding status at 4.5 to 6 months postpartum.

Results

A total of 644 mother-child pairs attended the baseline examination at birth (Fig 1).

Preterm births (n = 10) and mother–child pairs not residing in the city of Jimma who had only been to the hospital to give birth (n = 63) were excluded. Of the 571 children included in the follow-up study, 507 were eligible to be included in the growth modelling. Of these, 46, 56, 122, and 283 children had their FM and FFM assessed 3, 4, 5, and 6 times, respectively, during



Fig 1. Flow diagram of the study participants and number of fat mass and fat-free mass observations at each follow-up visit from birth to 6 months.

https://doi.org/10.1371/journal.pmed.1002888.g001

the first 6 months of follow-up (median number of measurements: 6). Thus, the LSME growth modelling included a total of 2,670 assessments of FM and FFM. The subsequent regression analyses included 340 children as 166 were not able to attend the 5-year visit or were lost to follow-up and 1 child did not have complete data on all covariates. Background characteristics, BC measures, and cardiometabolic markers of the mother–child pairs included in the regression analyses are presented in Tables 1 and 2.

Characteristic	Full sample $(n = 340)$	Girls (<i>n</i> = 171)	Boys (<i>n</i> = 169)	<i>p</i> -Value ²	Missing, n
Maternal characteristics					
Age at birth (years)	24.5 (4.7), 15.0 to 38.0	24.9 (4.8), 15.0 to 38.0	24.1 (4.5), 16.0 to 37.0	0.117	0
Postpartum height (cm)	157.1 (6.1), 142.0 to 174.0	157.5 (6.3), 143.5 to 174.0 156.6 (5.9), 142.0 to 173.6		0.197	0
Postpartum body mass index (kg/m ²)	22.20 (3.52), 14.42 to 36.32	22.22 (3.43), 14.42 to 36.32	22.18 (3.61), 15.85 to 34.21	0.919	3
Birth order of current child					
First	168 (49.4)	78 (45.6)	90 (53.3)		
Second	91 (26.8)	45 (26.3)	46 (27.2)		
Third or above	81 (23.8)	48 (28.1)	33 (19.5)	0.162	0
Breastfeeding status at 4.5 to 6 months postpartum					
Exclusive	41 (12.5)	20 (12.0)	21 (13.0)		
Almost exclusive (water given)	70 (21.4)	34 (20.5) 36 (22.4)			
Predominant	198 (60.6)	102 (61.4)	96 (59.6)		_
Partial/none	18 (5.5)	10 (6.0)			13
Maternal education				0.938	
No school	24 (7.1)	10 (5.8)	14 (8.3)		
Some primary school	153 (45.0)	78 (45.6)			
Completed primary school	54 (15.9)	36 (21.1)	18 (10.7)		
Completed secondary school	65 (19.1)	26 (15.2)	39 (23.1)		
Higher education	44 (12.9)	21 (12.3)	23 (13.6)		0
ocioeconomic status (International Wealth Index)	45.5 (17.1), 8.1 to 93.5	46.2 (17.5), 8.1 to 93.5	44.8 (16.7), 10.4 to 93.5	0.052	0
Child characteristics at birth					
Gestational age (weeks)	39.0 (1.0), 37.0 to 42.0	39.1 (1.0), 37.0 to 42.0	39.0 (0.9), 37.0 to 42.0	0.776	0
Veight (kg)	3.05 (0.40), 2.04 to 4.21	3.00 (0.41), 2.05 to 4.21			0
ength (cm)	49.2 (1.9), 43.0 to 54.5	48.9 (1.9), 43.5 to 53.8	49.4 (1.9), 43.0 to 54.5	0.027	0
fat mass (kg)	0.22 (0.16), -0.17 to 1.06	0.23 (0.16), -0.12 to 0.66			0
Fat-free mass (kg)	2.83 (0.32), 1.95 to 3.80	2.77 (0.31), 1.95 to 3.54	2.89 (0.32), 2.07 to 3.80	0.171	0
low birth weight ³	31 (9.1)	18 (10.5)	13 (7.7)	0.472	0
Child characteristics at 5 years				0.172	
Age at 5-year visit (months)	59.98 (1.41), 51.29 to 65.12	59.95 (1.58), 51.29 to 65.12	60.01 (1.21), 54.47 to 63.57	0.725	0
Veight (kg)	16.27 (2.06), 11.76 to 26.00	16.15 (2.03), 11.76 to 25.69	16.40 (2.09), 12.52 to 26.00	0.254	0
Height (cm)	104.2 (4.4), 91.5 to 115.0	104.0 (4.3), 93.6 to 114.0	104.3 (4.6), 91.5 to 115.0	0.536	0
Body mass index (kg/m ²)	14.96 (1.20), 11.47 to 20.89	14.89 (1.30), 11.47 to 20.89	15.03 (1.09), 12.41 to 19.66	0.312	0
Naist circumference (cm)	51.4 (3.0), 43.0 to 62.8	51.1 (3.1), 43.0 to 62.8	51.7 (2.9), 45.2 to 59.0	0.098	1
Veight for age $(z$ -score) ⁴	-0.90 (0.87), -3.18 to 2.37	-0.90 (0.83), -3.18 to 2.15			0
Height for age $(z$ -score) ⁴	-1.17 (0.91), -3.76 to 1.19	-1.12 (0.86), -3.23 to 1.19	-1.22 (0.96), -3.76 to 0.87	0.960	0
BMI for age $(z$ -score) ⁴	-0.25 (0.87), -3.16 to 2.82	-0.32 (0.90), -3.16 to 2.82	-0.19 (0.83), -2.50 to 2.72	0.152	0
Jnderweight ⁵	32 (9.4)	15 (8.8)	17 (10.1)	0.132	0
tunted ⁶	53 (15.6)	24 (14.0)	29 (17.2)	0.823	0
Vasted by BMI (thinness) ⁷	10 (2.9)	8 (4.7)	29(17.2)	0.519	0
Dverweight ⁸	14 (4.1)	7 (4.1)	7 (4.1)	1.000	0
Dese ⁹	4 (1.2)	2 (1.2)	2 (1.2)	1.000	0

Table 1. Description of the mother-child pairs included in the modelling of fat mass and fat-free mass velocity and attending the 5-year follow-up visit¹.

¹Data are mean (SD), minimum to maximum range, for continuous, normally distributed variables and count (%) for categorical variables.

²Differences between girls and boys were calculated by 1-way ANOVA F-test for continuous variables, Pearson's chi-squared test of independence for categorical variables with expected counts \geq 5 in all cells, and Fisher's exact test of independence for categorical variables with expected count in any cell < 5.

 ^{3}Low birth weight is defined as birth weight < 2,500 g.

PLOS MEDICINE

 ^{4}z -Scores are derived using the 2006 (children aged < 61 months) and 2007 (children aged ≥ 61 months) World Health Organization (WHO) child growth standards.

⁵Weight for age more than 2 SD below the sex-specific median of the WHO child growth standards.

⁶Height for age more than 2 SD below the sex-specific median of the WHO child growth standards.

⁷BMI for age more than 2 SD below the sex-specific median of the WHO child growth standards.

⁸BMI for age from 1 to 2 SD above the sex-specific median of the WHO child growth standards [40].

⁹BMI for age more than 2 SD above the sex-specific median of the WHO child growth standards [40].

https://doi.org/10.1371/journal.pmed.1002888.t001

	MEDICINE
--	----------

Outcome	Full sample (<i>n</i> = 340)	Girls (n = 171)	Boys (<i>n</i> = 169)	<i>p</i> -Value ²	Missing, n
Glucose homeostasis					
Whole blood glucose (mmol/l)	5.89 (0.84), 3.80-9.90	5.86 (0.76), 3.80-8.40	5.93 (0.91), 4.10-9.90	0.444	23
HbA1c (mmol/mol)	38 (4), 27–62	38 (4), 29–55	37 (4), 27–62	0.479	79
Insulin (µU/ml) ³	5.98 (3.20-11.20), 0.20-45.99	7.24 (4.06–12.83), 0.20–45.99	5.35 (2.96-9.38), 0.20-33.33	0.005	31
C-peptide (ng/ml) ³	1.06 (0.65–1.51), 0.14–5.56	1.14 (0.75–1.69), 0.21–5.56	0.92 (0.56-1.41), 0.14-4.46	0.003	36
HOMA-IR ^{3,4}	1.29 (0.66-2.47), 0.03-12.21	1.56 (0.85-2.66), 0.03-12.21	1.11 (0.61–2.09), 0.05–8.32	0.008	31
Lipids					
Total cholesterol (mmol/l)	3.41 (0.61), 2.04–5.83	3.45 (0.64), 2.04–5.83	3.38 (0.58), 2.19-5.16	0.345	27
LDL cholesterol (mmol/l)	1.65 (0.56), 0.00-3.72	1.69 (0.58), 0.33-3.72	1.61 (0.55), 0.00-3.40	0.180	28
HDL cholesterol (mmol/l)	0.79 (0.26), 0.09–1.53	0.78 (0.27), 0.11–1.53	0.80 (0.24), 0.09-1.50	0.80 (0.24), 0.09–1.50 0.403	
Triglycerides(mmol/l) ³	0.95 (0.73-1.28), 0.38-3.75	0.93 (0.76-1.28), 0.44-3.75	0.97 (0.71-1.28), 0.38-3.52	0.721	32
Blood pressure					
Systolic (mm Hg)	87.6 (7.2), 70.0–110.0	87.9 (7.1), 70.0–110.0	87.3 (7.4), 70.0–110.0	87.3 (7.4), 70.0–110.0 0.446	
Diastolic (mm Hg)	54.2 (8.5), 40.0-80.0	54.5 (8.4), 40.0-80.0	54.0 (8.5), 40.0-70.0	0.547	2
Body composition					
Fat mass (kg)	4.14 (1.27), 1.17–9.83	4.11 (1.37), 1.17–9.83	4.16 (1.17), 1.48-8.41	0.715	16
Fat-free mass (kg)	12.14 (1.40), 9.29–17.99	12.02 (1.30), 9.34–17.73	12.26 (1.50), 9.29-17.99	0.119	16
Fat mass index (kg/m ²)	3.80 (1.07), 1.12–7.99	3.79 (1.17), 1.12–7.99	3.80 (0.96), 1.45-6.36	.45-6.36 0.881	
Fat-free mass index (kg/m ²)	11.18 (0.86), 8.96–13.96	11.12 (0.88), 8.96–13.64	11.25 (0.84), 9.39–13.96	0.187	16

Table 2. Cardiometabolic markers and body composition at 5 years of age in the children included in the modelling of fat mass and fat-free mass velocity and attending the 5-year follow-up visit¹.

¹Data are mean (SD), minimum–maximum range, for continuous variables that are normally distributed and median (interquartile range), minimum–maximum range, for continuous variables that do not follow a normal distribution.

²Differences between groups were calculated by 1-way ANOVA F-test for continuous, normally distributed variables. Variables found not to follow a normal distribution were log-transformed prior to the tests of group differences.

³Non-normally distributed.

⁴Homeostasis model assessment of insulin resistance (HOMA-IR) was calculated as insulin (μ U/ml) × glucose (mmol/l)/22.5.

https://doi.org/10.1371/journal.pmed.1002888.t002

The vast majority of mothers were either exclusively or predominantly breastfeeding at 4.5 to 6 months postpartum, and 48% had completed primary school or a higher level of education. The average wealth status of the families was slightly lower than the national urban average, but considerably higher than the national rural average (46/100 versus 52/100 [urban] and 12/100 [rural]) [26,41]. At birth, 9% were born with low birth weight, which is lower than the average for sub-Saharan Africa of 14% [42]. Compared to the WHO international growth standards [43], at 5 years, the children had on average a lower weight, height, and BMI for age. Stunting at 5 years was seen in 15.5% of the children, which is similar to the proportion in the capital, Addis Ababa, but markedly lower than the national urban average of 25% [44].

FM and FFM accretion in early infancy

The LSME models estimated FM and FFM at birth and FM and FFM growth velocity over the periods 0–3 and 3–6 months. Examples of estimated growth velocity curves for 3 selected children are presented in Fig 2.

During the first 6 months, the children had a much larger variation in FM growth velocity compared to FFM growth velocity (S1 and S2 Figs). The FM growth velocity from 0 to 3 months was on average 4.4 times greater than that from 3 to 6 months. Although the estimated FM was only 8% of the FFM component at birth, the average FM growth velocity from 0 to 3



Age in months

Fig 2. Growth velocity modelling of fat mass and fat-free mass. Estimated child-specific fat mass and fat-free mass growth velocity curves for 3 selected children (coloured curves) and average fat mass and fat-free mass growth velocity curves for the whole study sample (black curves), estimated from linear-spline mixed-effects modelling. The coloured points show the observed fat mass and fat-free mass measurements for each child. The vertical grey line shows the selected knot point at 3 months, and the slopes on each side of the knot point are the estimated growth velocities for each child. The graphs show how the estimated fat mass and fat-free mass at birth (intercept) and growth velocities (the slope parameters for fat mass and fat-free mass) may vary between children.

https://doi.org/10.1371/journal.pmed.1002888.g002

months was similar to that of FFM of 0.5 kg per month. This resulted in a considerable relative catch-up in FM compared to FFM in the first 3 months, and at 6 months the estimated FM was on average only 2.6 times smaller than FFM (Table 3).

Correlations between FM at birth and FM growth velocities as well as FFM at birth and FFM growth velocities are shown in pairs plots in <u>S3 Fig</u>. Matrices of the model assumption tests of the LSME modelling of FM and FFM growth velocity are shown in <u>S4 Fig</u>.

Associations of FM and FFM accretion in infancy with BC and cardiometabolic markers at 5 years

Associations of estimated FM and FFM at birth and FM and FFM growth velocity in the periods 0–3 and 3–6 months with BC and cardiometabolic markers at 5 years are presented in Fig 3. Tabular presentations of the results of the regression analyses are shown in <u>S1</u> and <u>S2</u> Tables.

Higher FFM, but not FM, at birth and higher FM and FFM accretion in the periods 0–3 and 3–6 months were associated with higher FM at 5 years. For instance, independent of

Body composition	Full sample (<i>n</i> = 507)	Girls	Boys (<i>n</i> = 257)	<i>p</i> -Value ²	Fomon's reference child ³	
		(<i>n</i> = 250)			Girls	Boys
Fat mass						
Estimated fat mass at birth (g)	217 (60)	213 (62)	220 (58)	0.156	495	486
Fat mass velocity, 0-3 months (g/month)	506 (138)	503 (136)	509 (140)	0.639	290	336
Fat mass velocity, 3-6 months (g/month)	116 (104)	122 (103)	109 (105)	0.162	183	181
Estimated fat mass at 6 months (g)	2,081 (613)	2,088 (595)	2,074 (631)	0.799	1,915	2,037
Observed fat mass at 60 months $(g)^4$	4,142 (1,275)	4,113 (1,369)	4,172 (1,172)	0.673	2,949	2,720
Fat-free mass						
Estimated fat-free mass at birth (g)	2,828 (286)	2,770 (290)	2,885 (271)	< 0.001	2,830	3,059
Fat-free mass velocity, 0-3 months (g/month)	516 (77)	491 (67)	540 (79)	< 0.001	516	627
Fat-free mass velocity, 3–6 months (g/month)	346 (68)	336 (65)	357 (70)	0.001	319	351
Estimated fat-free mass at 6 months (g)	5,414 (540)	5,250 (511)	5,573 (519)	< 0.001	5,335	5,993
Observed fat-free mass at 60 months (g) ⁴	12,137 (1,403)	12,021 (1,302)	12,258 (1,496)	0.129	14,711	15,950

Table 3. Average estimated fat mass and fat-free mass at birth, fat mass and fat-free mass velocities from birth to 6 months of age estimated with linear-spline mixed-effects modelling, and estimated fat mass and fat-free mass at 6 months in Ethiopian children¹.

¹Data are mean (SD).

²Differences between the Ethiopian boys and girls were calculated by 1-way ANOVA F-test.

³Fat mass and fat-free mass velocity (g/month) were calculated by the difference in mean fat mass and fat-free mass in grams at the end of the age interval and at the beginning of the age interval divided by the length in months of the age interval using data from the 1982 Fomon reference child [23].

 ^{4}n for full sample = 325.

https://doi.org/10.1371/journal.pmed.1002888.t003

current FFM and other covariates, 100-g/month higher FM accretion in the periods 0-3 and 3–6 months was associated with 339 g (95% CI: 243–435 g, p < 0.001) and 367 g (95% CI: 250– 484 g, p < 0.001) greater FM at 5 years, respectively (S2 Table). Higher FM and FFM at birth and accretion from 0-3 months were associated with higher FFM at 5 years. FFM accretion from 3 to 6 months was also associated with FFM at 5 years. For instance, independent of current FM and the other covariates, 100-g higher FFM at birth was associated with 189 g (95% CI: 134–243 g, p < 0.001) greater FFM, and in the periods 0–3 and 3–6 months, 100-g/month higher FFM accretion was associated with 1,002 g (95% CI: 815-1,189 g, p < 0.001) and 624 g (95% CI: 419–829 g, p < 0.001) greater FFM at 5 years, respectively. Additionally, FFM at birth and accretion in the periods 0-3 and 3-6 months of age as well as FM accretion in the period 0-3 months were positively associated with height at 5 years. For instance, 100-g/month higher FFM accretion in the period 0–3 months was associated with 1.8 cm (95% CI: 1.3–2.4 cm, p <0.001) greater height at 5 years, and 100-g/month higher FM accretion in the same period with 0.4 cm (95% CI: 0.1–0.7, p = 0.016) greater height. FM velocity in the periods 0–3 and 3–6 months, and FFM at birth and velocity from 0-3 months, were positively associated with waist circumference at 5 years.

Higher FM at birth and accretion from 0 to 3 months was associated with higher concentrations of total, LDL-, and HDL-cholesterol, and associations were strongest for LDL-cholesterol. For instance, independent of current FM and the other covariates, 100-g higher FM at birth was associated with 0.16 mmol/l (95% CI: 0.05–0.26 mmol/l, p = 0.005) greater LDL-cholesterol at 5 years, and 100-g/month greater FM accretion in the period 0–3 months with 0.06 mmol/l (95% CI: 0.01–0.12 mmol/l, p = 0.016) greater LDL-cholesterol (S2 Table). Neither FFM at birth nor accretion over the periods 0–3 and 3–6 months was associated with cholesterol concentrations at 5 years.

In general, the reported associations did not change markedly from model 1 to model 2. The presented associations also did not change markedly after adjusting for breastfeeding



Fig 3. Associations of estimated fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity over the periods 0–3 months and 3–6 months with body composition and cardiometabolic markers at 5 years. The coefficients (and 95% CIs) displayed in the forest plots were derived from separate multiple linear regression analyses and represent the change in the 5-year outcomes per study population standard deviation increase of estimated fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity over the periods 0–3 months and 3–6 months. The linear-spline mixed-effects model used to derive the estimated values at birth and growth velocities had 1 knot point at 3 months, yielding the 2 growth periods 0–3 and 3–6 months. Variables found not to follow a normal distribution (i.e., insulin, C-peptide, HOMA-IR, and triglycerides) were log-transformed prior to the regression analyses. The presented estimates for these variables were back-transformed and are shown as percentwise change. Model 1 was adjusted for child's sex, birth order, gestational age at birth, child's exact age at the 5-year visit, maternal age at delivery, maternal postpartum height, maternal educational status, and family socioeconomic status (International Wealth Index). Model 2 was additionally adjusted for fat mass at the 5-year visit (applies to all outcomes except fat mass and waist circumference, which were adjusted for fat-free mass at the 5-year visit instead of fat mass in model 2). *p < 0.05, **p < 0.01, ***p < 0.001.

https://doi.org/10.1371/journal.pmed.1002888.g003

status at 4.5 to 6 months postpartum (S5 Fig). When accounting for multiple testing in model 2, the associations for the markers of lipid metabolism were no longer significant except for the association of FM at birth with LDL-cholesterol at 5 years (S6 Fig). The associations of FM and FFM at birth and their accretion in infancy with height, waist circumference, FM, and FFM largely remained significant in the model 2 analyses. We found no associations of FM and FFM at birth and their accretion in infancy with any of the other studied cardiometabolic markers including glucose, HbA1c, insulin, C-peptide, HOMA-IR, triglycerides, and systolic and diastolic blood pressure.

Discussion

In this contemporary cohort of urban Ethiopian children, we found that FM accretion in infancy was positively associated with markers of adiposity (FM and waist circumference) at 5 years, and that higher FFM at birth and accretion in infancy were associated with FFM, FM, height, and waist circumference at 5 years. In addition, we found that higher FM accretion during fetal life, indexed by FM at birth, and higher FM accretion from 0 to 3 months predicted higher cholesterol concentrations at 5 years, with the strongest association for LDL-cholesterol. However, none of the other studied cardiometabolic markers were associated with FM or FFM at birth or their accretion during infancy.

In high-income populations, fetal and postnatal growth have consistently been related to indices of obesity and markers of cardiometabolic disease risk [5,6,45]. These findings have been crucial for the identification of critical windows in early life in which later obesity and cardiometabolic disease risk are being programmed. However, previous studies have treated the body as a single component by using weight or BMI as the index of growth, and have therefore not been able to assess how the metabolically diverse body components of FM and FFM might mediate associations of growth in early life with later BC and cardiometabolic risk markers. There is also less information on such associations in low- and middle-income populations, though data from India support the notion that fetal life and infancy are key critical windows [34,46,47]. We have previously shown that greater FM accretion from 0 to 4 months predicted higher FM at age 4 years, and that greater FFM accretion from 0 to 6 months predicted higher FM and FFM at 4 years of age [32]. To our knowledge, no other cohort studies have examined the associations of directly assessed FM and FFM accretion during infancy with BC and cardiometabolic risk markers in early childhood. Using a 3-way conditional growth analysis of weight, height, and skinfolds to approximate fat and lean tissue gain in discrete age intervals from birth to 13.5 years, Krishnaveni et al. [33] found that faster fat accretion from 5 to 9.5 years predicted higher fat percentage, waist-hip ratio, systolic blood pressure, insulin, and HOMA-IR at 13.5 years, but they reported no associations of fat or lean tissue at birth and growth over the periods 0-1 and 1-2 years with later BC or any of the studied cardiometabolic outcomes. In a small observational study, Koontz et al. [48] found that accelerated FM accretion from 0 to 8 months of age was associated with an 8-fold increase in the odds of overweight/obesity at 9 years. Previous studies have used changes in weight and/or height as indicators of growth and have found birth weight and weight gain in infancy to be associated with later FM and overweight/obesity [48-50], FFM [37,51], or both FM and FFM [52,53].

The mechanisms relating higher FFM accretion in infancy to higher FFM in childhood and higher FM accretion in infancy to higher FM in childhood remain elusive. Besides cellular and extracellular water, FFM consists primarily of protein followed by osseous minerals [23]. In a large birth cohort study from the Netherlands, higher fetal and postnatal growth in weight and height predicted higher bone mineral content, density, and area [54]. The strongest

associations for the bone outcomes were seen for growth in the first year. Greater FFM accretion in early infancy may therefore promote bone growth, likely through a complex interplay with growth hormone and insulin-like growth factor 1 (IGF-1) [55], which in turn contributes to higher FFM later. In addition, the accretion of muscle mass is likely to be a beneficial investment for later cardiometabolic health. Even though FFM accretion was not related to markers of glucose homeostasis at 5 years in the present study, muscle is an important factor for the uptake of glucose from the blood in response to insulin secretion, and low FFM may therefore influence later insulin sensitivity [51].

Regarding the cardiometabolic risk markers, we found that FM at birth and accretion from 0 to 3 months were positively associated with cholesterol concentrations at 5 years, with the largest estimates for LDL-cholesterol. Interestingly, the associations remained after controlling for current FM. While FM is important for brain development and resilience to infections in early life [56], excessive FM accretion in infancy may also lead to harmful levels of FM and a perturbed lipid metabolism later in childhood. As shown in Table 3, compared to reference data for child BC developed by Fomon et al. [23], both boys and girls in the present study had less than half as much absolute FM at birth. At 6 months of age, absolute FM was slightly higher for the Ethiopian boys and around 175 g higher for girls. Surprisingly, relative to the Fomon reference values, the Ethiopian boys and girls had over 1.4 and 1.2 kg higher FM, but a deficit in FFM of 3.7 and 2.7 kg, respectively, at 5 years of age. Additionally, compared to BC reference data from the UK, the Ethiopian children also had higher FM (boys: 1.1 kg higher; girls: 0.2 kg higher) and markedly lower FFM (boys: 4.1 kg lower; girls: 2.6 kg lower) at 5 years [57]. In fact, even compared to 6-year-old children from Pune in India [34], already at 5 years of age, the Ethiopian children had a higher absolute FM (boys: 1.3 kg higher; girls: 0.8 kg higher) and lower absolute FFM (boys: 0.7 kg lower; girls: 0.1 kg lower). The Indian children have been described as having a 'thin-fat' phenotype, because of their low weight at birth, short stature, and relatively high FM at birth and in childhood, and have been recognised as being particularly vulnerable to developing obesity and cardiometabolic diseases later in life [58]. In this Ethiopian population, we see an even more pronounced thin-fat phenotype at 5 years, resulting from a considerable catch-up in FM, and this may explain some of the variation we see in the cholesterol concentrations at 5 years. Only a few children presented lipid concentrations exceeding the cutoffs for high total and LDL-cholesterol [59], and at this age we cannot say whether increases in total and LDL-cholesterol associated with FM accretion in infancy have health implication in the longer term or rather represent beneficial metabolic adaptations to greater tissue mass. However, cardiometabolic risk markers have been found to track from childhood to adulthood [60,61], and adiposity in childhood has been associated with risk of obesity and cardiometabolic disease later in life [62-64]. Furthermore, even small increases in LDL-cholesterol could potentially promote atherosclerotic processes already in childhood, as the probability of intimal retention of LDL-cholesterol particles, and hence the risk of developing atherosclerotic plaques, increases in a dose-response relationship when circulating concentrations of LDL-cholesterol exceed concentrations as low as 0.5-1.0 mmol/l [65]. Furthermore, the rate of atherosclerotic plaque formation is further increased with risk factors such as genetic predisposition, high blood pressure, high-fat diets, and adiposity [66]. On average, the HDL-cholesterol concentration was 0.79 mmol/l, which is well below the recommendation for children that HDL-cholesterol concentration should be above 1.16 mmol/l [59]. Thus, a low average birth weight, a high level of FM, a low level of FFM, and a low concentration of HDL-cholesterol at school age as well as exposure to a rapidly changing obesogenic food environment could potentially interact with LDL-cholesterol concentrations and make Ethiopian children and similar populations particularly vulnerable to later cardiometabolic disease. Surprisingly, we also found that FM at birth and accretion from 0 to 3 months

PLOS Medicine | https://doi.org/10.1371/journal.pmed.1002888 August 20, 2019

were associated with greater HDL-cholesterol, which could suggest general high turnover of apolipoproteins with adipose tissue accumulation.

Strengths and limitations

A major strength of this study was the detailed assessment of FM and FFM from birth to 6 months and again at 60 months, using accurate and validated equipment for estimation of BC. A high density of BC data during the first 6 months allowed us to robustly estimate the nonlinear relationship of FM and FFM as a function of age in critical windows of development in early childhood using LSME modelling. This modelling approach provided easily interpretable growth coefficients from a complex data structure of repeated measures, and at the same time accounted for the dependencies of the child-specific BC measurements. Our growth models also allowed children to be included in the modelling even if they were not measured at the exact same time or had not attended all 6 follow-up visits during infancy. Thus, as opposed to more conventional growth modelling approaches, such as conditional growth modelling or tracing average z-scores, that require a complete case analysis, we were able to reduce potential bias from selective dropout and increase the statistical power of the study considerably. This novel approach to growth modelling has been implemented successfully in previous studies [28,67,68]. Finally, of the children initially included in the follow-up study (n = 571), 89% were included in the modelling of FM and FFM accretion in infancy, and 60% of the mother-child pairs were able to participate in the follow-up visit at 5 years, where data on the cardiometabolic and BC outcomes were collected.

Among the limitations, we cannot rule out that the loss of mother-child pairs examined at birth but not able to attend the visit at 5 years could have introduced selection bias, although mother-child pairs lost to follow-up were generally similar to those included in the 5-year analyses (S3 Table). Also, as the PEA POD was only able to measure children from birth to 6 months of age, we did not have FM and FFM assessments in the period from 6 months to 4 years. Previous studies have found that weight gain in early childhood, rather than infancy and fetal life, is the strongest correlate of concentrations of insulin and C-peptide and of HOMA-IR [31,34]. Thus, information on FM and FFM accretion from 6 months to 4 years could have provided valuable insight into the relative importance of FM and FFM accretion beyond infancy. Additionally, as it was not feasible for the small children to undergo an overnight fast, the fasting time was standardised to 3 hours. This may have caused some non-differential misclassification of the estimates related to glucose homeostasis (i.e., estimates moving towards 0). However, for the lipid measurements, as most children are in a non-fasting state most of the day, the concentrations observed after 3 hours of fasting are likely to be a better reflection of the lipid and lipoprotein concentrations in the circulating blood than after an overnight fast [69]. The advantages of using non-fasting measurements of daily lipid and lipoprotein concentrations have resulted in recommendations to include non-fasting lipid measurements in several national society guidelines, including the Danish Society of Clinical Biochemistry and the National Institute for Health and Care Excellence (NICE) in the UK. Finally, the observational study design precludes us from making any claims of causal inference, and we can therefore not exclude the possibility that residual confounding from important covariates such as pre-pregnancy maternal nutritional status, gestational weight gain, paternal BMI, fetal growth trajectories, diet in infancy and childhood, and duration of breastfeeding could have resulted in the identified associations. A longer duration of breastfeeding is inversely associated with the risk of overweight [70]. Moreover, mode of infant feeding is associated with differences in early-life accretion of FM and FFM, with lower FM accretion from birth to 6 months among formula-fed compared to breastfed infants but higher FM at 12

months [71]. Correspondingly, in the present cohort, we have previously shown that not being breastfed at 3 months of age was associated with a delayed FM accretion pattern [14]. Evidence on the cardiometabolic effects of breastfeeding remains controversial, with some studies suggesting a beneficial effect of breastfeeding in reducing adiposity and cardiometabolic risk [72–74] and others finding no to little positive effect on various cardiometabolic outcomes [75–78]. Our breastfeeding data were crude, and unfortunately we did not have data on complementary feeding. Nevertheless, in a sensitivity analysis, we adjusted our main analysis for breastfeeding status at 4.5 to 6 months postpartum, which did not affect the identified associations markedly (S5 Fig). Furthermore, the high levels of predominant and exclusive breastfeeding in this population suggest that breastfeeding may not have been the main source of variability.

Our findings suggest that development of adiposity and changes in cardiometabolic risk markers related to lipid metabolism may occur already in early life and are likely to be driven by accelerated fetal and early postnatal FM accretion. In contrast, FFM accretion in early life is likely to improve cardiometabolic status through linear growth and continued accretion of lean tissue, even though we were not yet able to demonstrate any direct benefits in the cardiometabolic markers. Thus, based on existing evidence from the iABC cohort and the present findings, we stress the importance of promoting lean tissue accretion and linear growth without excess accumulation of adipose tissue in countries undergoing rapid economic and nutritional transitions, as children in such contexts are likely to be particularly exposed to an environment conducive to an unhealthy lifestyle. This may be done through public health policies that promote healthy diets and physical activity in preschools and schools as well as limit the exposure to obesogenic environments (outlets with soft drinks, fast food, and unhealthy snacks) within schools and in the community [79]. Lastly, promotion of exclusive breastfeeding as the most optimal nutrition for the child is key to ensure normal growth, prevent infections, and reduce the risk obesity in childhood and adolescence [78].

Supporting information

S1 Fig. Growth velocities from birth to 6 months of age for fat mass and fat-free mass estimated with linear-spline mixed-effects modelling. (PDF)

S2 Fig. Density plots of the variation of fat mass and fat-free mass growth velocity in the periods 0-3 and 3-6 months of age. (PDF)

S3 Fig. Correlation matrix (pairs plot) of the child-specific standard deviation (SD) scores of estimated fat mass and fat-free mass at birth and SD scores of fat mass and fat-free mass growth velocities in the periods 0–3 and 3–6 months. (PDF)

S4 Fig. Matrix of model assumption tests of the linear-spline mixed-effects modelling of fat mass and fat-free mass growth velocity. (PDF)

S5 Fig. Sensitivity analyses of the associations shown in Fig 3. Associations of estimated fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity in the periods 0–3 and 3–6 months with body composition and cardiometabolic risk markers at 5 years (all analyses adjusted for breastfeeding at 4.5 to 6 months postpartum). (PDF)

S6 Fig. Sensitivity analyses accounting for multiple testing in model 2 of the associations of estimated fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity for the periods 0–3 and 3–6 months with body composition and cardiometabolic risk markers at 5 years.

(PDF)

S1 Table. Associations of predicted fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity in the periods 0–3 and 3–6 months with cardiometabolic markers and body composition at 5 years in the fully adjusted model 2 (exposures in SD units). (PDF)

S2 Table. Associations of predicted fat mass and fat-free mass at birth and fat mass and fat-free mass growth velocity in the periods 0–3 and 3–6 months with cardiometabolic markers and body composition at 5 years in the fully adjusted model 2 (exposures in absolute values).

(PDF)

S3 Table. Comparison of background characteristics of the mother-child pairs attending the 5-year follow-up visit with those not attending. (PDF)

S1 Text. STROBE statement. (PDF)

S2 Text. Study protocol: Body composition trajectories and metabolic risk in African children.

(PDF)

S3 Text. Detailed description of the linear-spline mixed-effects modelling. (PDF)

Acknowledgments

We are thankful to the research and administrative staff of the Jimma University Clinical and Nutrition Research Partnership (JUCAN) for their strong commitment to this study. Furthermore, we thank the women and children for their participation throughout the years. We also thank the Ethiopian Public Health Institute for the laboratory analyses of our serum samples.

Author Contributions

- **Conceptualization:** Tsinuel Girma, Bitiya Admassu, Mubarek Abera, Pernille Kæstel, Kim F. Michaelsen, Henrik Friis, Jonathan C. K. Wells, Gregers S. Andersen.
- **Data curation:** Rasmus Wibaek, Bitiya Admassu, Mubarek Abera, Alemseged Abdissa, Pernille Kæstel, Gregers S. Andersen.
- Formal analysis: Rasmus Wibaek, Dorte Vistisen.
- **Funding acquisition:** Rasmus Wibaek, Tsinuel Girma, Marit E. Jørgensen, Pernille Kæstel, Kim F. Michaelsen, Henrik Friis, Jonathan C. K. Wells, Gregers S. Andersen.

Investigation: Rasmus Wibaek, Pernille Kæstel, Gregers S. Andersen.

Methodology: Rasmus Wibaek, Dorte Vistisen, Marit E. Jørgensen, Pernille Kæstel, Jonathan C. K. Wells.

Project administration: Rasmus Wibaek, Tsinuel Girma, Bitiya Admassu, Mubarek Abera, Alemseged Abdissa, Pernille Kæstel, Gregers S. Andersen.

Resources: Rasmus Wibaek.

Software: Rasmus Wibaek.

Supervision: Tsinuel Girma, Marit E. Jørgensen, Pernille Kæstel, Kim F. Michaelsen, Henrik Friis, Jonathan C. K. Wells, Gregers S. Andersen.

Visualization: Rasmus Wibaek.

Writing - original draft: Rasmus Wibaek.

Writing – review & editing: Rasmus Wibaek, Dorte Vistisen, Tsinuel Girma, Bitiya Admassu, Mubarek Abera, Alemseged Abdissa, Marit E. Jørgensen, Pernille Kæstel, Kim F. Michaelsen, Henrik Friis, Jonathan C. K. Wells, Gregers S. Andersen.

References

- Mortality and Causes of Death Collaborators. Global, regional, and national life expectancy, all-cause mortality, and cause-specific mortality for 249 causes of death, 1980–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016; 388:1459–544. https://doi.org/10.1016/ S0140-6736(16)31012-1 PMID: 27733281
- GBD 2015 DALYs and HALE Collaborators. Global, regional, and national disability-adjusted life-years (DALYs) for 315 diseases and injuries and healthy life expectancy (HALE), 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016; 388:1603–58. https://doi.org/10. 1016/S0140-6736(16)31460-X PMID: 27733283
- Whincup PH, Kaye SJ, Owen CG, Huxley R, Cook DG, Anazawa S, et al. Birth weight and risk of type 2 diabetes: a systematic review. JAMA. 2008; 300:2886–97. <u>https://doi.org/10.1001/jama.2008.886</u> PMID: 19109117
- Baird J, Fisher D, Lucas P, Kleijnen J, Roberts H, Law C. Being big or growing fast: systematic review of size and growth in infancy and later obesity. BMJ. 2005; 331:929–0. https://doi.org/10.1136/bmj.38586. 411273.E0 PMID: 16227306
- Eriksson JG. Early growth and coronary heart disease and type 2 diabetes: findings from the Helsinki Birth Cohort Study (HBCS). Am J Clin Nutr. 2011; 94:1799–802. <u>https://doi.org/10.3945/ajcn.110.</u> 000638 PMID: 21613556
- Barker DJP, Eriksson JG, Forsén T, Osmond C. Fetal origins of adult disease: strength of effects and biological basis. Int J Epidemiol. 2002; 31:1235–9. <u>https://doi.org/10.1093/ije/31.6.1235</u> PMID: 12540728
- Lucas A, Fewtrell MS, Cole TJ. Fetal origins of adult disease-the hypothesis revisited. BMJ. 1999; 319:245–9. https://doi.org/10.1136/bmj.319.7204.245 PMID: 10417093
- Singhal A, Lucas A. Early origins of cardiovascular disease: is there a unifying hypothesis? Lancet. 2004; 363:1642–5. https://doi.org/10.1016/S0140-6736(04)16210-7 PMID: 15145640
- Huxley RR, Shiell AW, Law CM. The role of size at birth and postnatal catch-up growth in determining systolic blood pressure: a systematic review of the literature. J Hypertens. 2000; 18:815–31. PMID: 10930178
- Ong KK, Dunger DB. Birth weight, infant growth and insulin resistance. Eur J Endocrinol. 2004; 151: U131–9. https://doi.org/10.1530/eje.0.151U131 PMID: 15554898
- 11. World Health Organization. Global status report on noncommunicable diseases 2010. Geneva: World Health Organization; 2011.
- Wells JCK, Fewtrell MS. Is body composition important for paediatricians? Arch Dis Child. 2008; 93:168–72. https://doi.org/10.1136/adc.2007.115741 PMID: 17804592
- Andersen GS, Girma T, Wells JCK, Kæstel P, Michaelsen KF, Friis H. Fat and fat-free mass at birth: air displacement plethysmography measurements on 350 Ethiopian newborns. Pediatr Res. 2011; 70:501–6. https://doi.org/10.1203/PDR.0b013e31822d7470 PMID: 21772228
- Andersen GS, Wibaek R, Kaestel P, Girma T, Admassu B, Abera M, et al. Body composition growth patterns in early infancy: a latent class trajectory analysis of the Ethiopian iABC birth cohort. Obesity (Silver Spring). 2018; 26:1225–33. https://doi.org/10.1002/oby.22197 PMID: 29845756

- Andersen GS, Girma T, Wells JC, Kæstel P, Leventi M, Hother AL, et al. Body composition from birth to 6 mo of age in Ethiopian infants: reference data obtained by air-displacement plethysmography. Am J Clin Nutr. 2013; 98:885–94. https://doi.org/10.3945/ajcn.113.063032 PMID: 23985805
- Central Statistical Agency. Statistical report on the 2013 national labour survey. Addis Ababa: Central Statistical Agency; 2014.
- Wells JCK, Fuller NJ, Wright A, Fewtrell MS, Cole TJ. Evaluation of air-displacement plethysmography in children aged 5–7 years using a three-component model of body composition. Br J Nutr. 2003; 90:699–707. https://doi.org/10.1079/bjn2003930 PMID: 13129477
- Fields DA, Allison DB. Air-displacement plethysmography pediatric option in 2–6 years old using the four-compartment model as a criterion method. Obesity (Silver Spring). 2012; 20:1732–7. <u>https://doi.org/10.1038/oby.2012.28 PMID: 22421895</u>
- Ellis KJ, Yao M, Shypailo RJ, Urlando A, Wong WW, Heird WC. Body-composition assessment in infancy: air-displacement plethysmography compared with a reference 4-compartment model. Am J Clin Nutr. 2007; 85:90–5. https://doi.org/10.1093/ajcn/85.1.90 PMID: 17209182
- Urlando A, Dempster P, Aitkens S. A new air displacement plethysmograph for the measurement of body composition in infants. Pediatr Res. 2003; 53:486–92. <u>https://doi.org/10.1203/01.PDR.</u> 0000049669.74793.E3 PMID: 12595599
- Dempster P, Aitkens S. A new air displacement method for the determination of human body composition. Med Sci Sports Exerc. 1995; 27:1692–7. PMID: 8614327
- Wells JCK, Fuller NJ, Dewit O, Fewtrell MS, Elia M, Cole TJ. Four-component model of body composition in children: density and hydration of fat-free mass and comparison with simpler models. Am J Clin Nutr. 1999; 69:904–12. https://doi.org/10.1093/ajcn/69.5.904 PMID: 10232629
- Fomon SJ, Haschke F, Ziegler EE, Nelson SE. Body composition of reference children from birth to age 10 years. Am J Clin Nutr. 1982; 35:1169–75. https://doi.org/10.1093/ajcn/35.5.1169 PMID: 7081099
- Matthews DR, Hosker JP, Rudenski AS, Naylor BA, Treacher DF, Turner RC. Homeostasis model assessment: insulin resistance and beta-cell function from fasting plasma glucose and insulin concentrations in man. Diabetologia. 1985; 28:412–9. PMID: 3899825
- Ballard JL, Novak KK, Driver M. A simplified score for assessment of fetal maturation of newly born infants. J Pediatr. 1979; 95:769–74. https://doi.org/10.1016/s0022-3476(79)80734-9 PMID: 490248
- Smits J, Steendijk R. The International Wealth Index (IWI). Soc Indic Res. 2015; 122:65–85. <u>https://doi.org/10.1007/s11205-014-0683-x</u>
- Labbok M, Krasovec K. Toward consistency in breastfeeding definitions. Stud Fam Plann. 1990; 21:226–30. PMID: 2219227
- Howe LD, Tilling K, Matijasevich A, Petherick ES, Santos AC, Fairley L, et al. Linear spline multilevel models for summarising childhood growth trajectories: a guide to their application using examples from five birth cohorts. Stat Methods Med Res. 2016; 25:1854–74. <u>https://doi.org/10.1177/</u> 0962280213503925 PMID: 24108269
- Tilling K, Macdonald-Wallis C, Lawlor DA, Hughes RA, Howe LD. Modelling childhood growth using fractional polynomials and linear splines. Ann Nutr Metab. 2014; 65:129–38. <u>https://doi.org/10.1159/</u> 000362695 PMID: 25413651
- Ekelund U, Ong KK, Linne Y, Neovius M, Brage S, Dunger DB, et al. Association of weight gain in infancy and early childhood with metabolic risk in young adults. J Clin Endocrinol Metab. 2007; 92:98– 103. https://doi.org/10.1210/jc.2006-1071 PMID: 17032722
- Voerman E, Jaddoe VW, Franco OH, Steegers EA, Gaillard R. Critical periods and growth patterns from fetal life onwards associated with childhood insulin levels. Diabetologia. 2017; 60:81–8. <u>https://doi.org/10.1007/s00125-016-4135-9 PMID: 27757489</u>
- Admassu B, Wells JCK, Girma T, Belachew T, Ritz C, Owino V, et al. Body composition during early infancy and its relation with body composition at 4 years of age in Jimma, an Ethiopian prospective cohort study. Nutr Diabetes. 2018; 8:46. https://doi.org/10.1038/s41387-018-0056-7 PMID: 30190452
- Krishnaveni GV, Veena SR, Srinivasan K, Osmond C, Fall CH. Linear growth and fat and lean tissue gain during childhood: associations with cardiometabolic and cognitive outcomes in adolescent Indian children. PLoS ONE. 2015; 10(11):e0143231. https://doi.org/10.1371/journal.pone.0143231 PMID: 26575994
- 34. Joglekar CV, Fall CHD, Deshpande VU, Joshi N, Bhalerao A, Solat V, et al. Newborn size, infant and childhood growth, and body composition and cardiovascular disease risk factors at the age of 6 years: the Pune Maternal Nutrition Study. Int J Obes (Lond). 2007; 31:1534–44. https://doi.org/10.1038/sj.ijo. 0803679 PMID: 17653070
- Hoffman DJ, Reynolds RM, Hardy DB. Developmental origins of health and disease: current knowledge and potential mechanisms. Nutr Rev. 2017; 75:951–70. https://doi.org/10.1093/nutrit/nux053 PMID: 29186623

- Reynolds RM, Osmond C, Phillips DI, Godfrey KM. Maternal BMI, parity, and pregnancy weight gain: influences on offspring adiposity in young adulthood. J Clin Endocrinol Metab. 2010; 95:5365–9. https://doi.org/10.1210/jc.2010-0697 PMID: 20702520
- Wells JC, Hallal PC, Wright A, Singhal A, Victora CG. Fetal, infant and childhood growth: relationships with body composition in Brazilian boys aged 9 years. Int J Obes (Lond). 2005; 29:1192–8. <u>https://doi.org/10.1038/sj.ijo.0803054</u> PMID: 16103893
- Wells JCK, Hallal PC, Reichert FF, Dumith SC, Menezes AM, Victora CG. Associations of birth order with early growth and adolescent height, body composition, and blood pressure: prospective birth cohort From Brazil. Am J Epidemiol. 2011; 174:1028–35. https://doi.org/10.1093/aje/kwr232 PMID: 21940799
- Benjamini Y, Hochberg Y. Controlling the false discovery rate: a practical and powerful approach to multiple testing. J R Stat Soc Series B Stat Methodol. 1995; 57:289–300.
- 40. NCD Risk Factor Collaboration (NCD-RisC). Worldwide trends in body-mass index, underweight, overweight, and obesity from 1975 to 2016: a pooled analysis of 2416 population-based measurement studies in 128-9 million children, adolescents, and adults. Lancet. 2017; 390:2627–42. https://doi.org/10. 1016/S0140-6736(17)32129-3 PMID: 29029897
- Global Data Lab. GDL Area Database (3.3.1). Nijmegen: Nijmegen Center for Economics; 2018 [cited 2018 Jun 13]. https://globaldatalab.org/areadata/.
- UNICEF Data. Low birthweight. Geneva: United Nations Children's Fund; 2019 [cited 2019 May 30]. https://data.unicef.org/topic/nutrition/low-birthweight/.
- 43. WHO Multicentre Growth Reference Study Group. WHO child growth standards: length/height-for-age, weight-for-length, weight-for-height and body mass index-for-age—methods and development. Geneva: World Health Organization; 2006.
- Central Statistical Agency. Ethiopia Demographic and Health Survey 2016. Addis Ababa: Central Statistics Agency; 2017.
- Cottrell EC, Ozanne SE. Early life programming of obesity and metabolic disease. Physiol Behav. 2008; 94:17–28. https://doi.org/10.1016/j.physbeh.2007.11.017 PMID: 18155097
- Bhargava SK, Sachdev HS, Fall C, Osmond C, Lakshmy R, Barker D, et al. Relation of serial changes in childhood body-mass index to impaired glucose tolerance in young adulthood. N Engl J Med. 2004; 350:865–75. https://doi.org/10.1056/NEJMoa035698 PMID: 14985484
- Yajnik CS, Fall C, Coyaji KJ, Hirve SS. Neonatal anthropometry: the thin–fat Indian baby. The Pune maternal nutrition study. Int J Obes Relat Metab Disord. 2003:173–80. https://doi.org/10.1038/sj.ijo. 802219 PMID: 12586996
- Koontz MB, Gunzler DD, Presley L, Catalano PM. Longitudinal changes in infant body composition: association with childhood obesity. Pediatr Obes. 2014; 9:e141–4. <u>https://doi.org/10.1111/ijpo.253</u> PMID: 25267097
- Ong K, Ahmed ML, Emmett PM, Preece MA. Association between postnatal catch-up growth and obesity in childhood: prospective cohort study. BMJ. 2000; 320:967–71. https://doi.org/10.1136/bmj.320. 7240.967 PMID: 10753147
- Adair LS, Fall CH, Osmond C, Stein AD, Martorell R, Ramirez-Zea M, et al. Associations of linear growth and relative weight gain during early life with adult health and human capital in countries of low and middle income: findings from five birth cohort studies. Lancet. 2013; 382:525–34. <u>https://doi.org/10.1016/S0140-6736(13)60103-8</u> PMID: 23541370
- Singhal A, Wells J, Cole TJ, Fewtrell M, Lucas A. Programming of lean body mass: a link between birth weight, obesity, and cardiovascular disease? Am J Clin Nutr. 2003; 77:726–30. <u>https://doi.org/10.1093/ajcn/77.3.726</u> PMID: 12600868
- 52. de Beer M, Vrijkotte TG, Fall CH, van Eijsden M, Osmond C, Gemke RJ. Associations of infant feeding and timing of linear growth and relative weight gain during early life with childhood body composition. Int J Obes (Lond). 2015; 39:586–92. https://doi.org/10.1038/ijo.2014.200 PMID: 25435256
- Chomtho S, Wells JC, Williams JE, Davies PS, Lucas A, Fewtrell MS. Infant growth and later body composition: evidence from the 4-component model. Am J Clin Nutr. 2008; 87:1776–84. <u>https://doi.org/10.1093/ajcn/87.6.1776 PMID: 18541568</u>
- 54. Heppe DHM, Medina-Gomez C, de Jongste JC, Raat H, Steegers EAP, Hofman A, et al. Fetal and childhood growth patterns associated with bone mass in school-age children: the Generation R Study. J Bone Miner Res. 2014; 29:2584–93. https://doi.org/10.1002/jbmr.2299 PMID: 24935163
- Velloso CP. Regulation of muscle mass by growth hormone and IGF-I. Br J Pharmacol. 2008; 154:557– 68. https://doi.org/10.1038/bjp.2008.153 PMID: 18500379
- Kuzawa CW. Adipose tissue in human infancy and childhood: an evolutionary perspective. Am J Phys Anthropol;1998;Suppl 27:177–209.

- Wells JCK, Williams JE, Chomtho S, Darch T, Grijalva-Eternod C, Kennedy K, et al. Body-composition reference data for simple and reference techniques and a 4-component model: a new UK reference child. Am J Clin Nutr. 2012; 96:1316–26. https://doi.org/10.3945/ajcn.112.036970 PMID: 23076617
- Bavdekar A, Yajnik CS, Fall CH, Bapat S, Pandit AN, Deshpande V, et al. Insulin resistance syndrome in 8-year-old Indian children: small at birth, big at 8 years, or both? Diabetes. 1999; 48:2422–9. https:// doi.org/10.2337/diabetes.48.12.2422 PMID: 10580432
- 59. Expert panel on integrated guidelines for cardiovascular health and risk reduction in children and adolescents. Expert panel on integrated guidelines for cardiovascular health and risk reduction in children and adolescents: summary report. Pediatrics. 2011; 128(Suppl 5):S213–56. https://doi.org/10.1542/ peds.2009-2107C PMID: 22084329
- 60. Hayman LL, Meininger JC, Daniels SR, McCrindle BW, Helden L, Ross J, et al. Primary prevention of cardiovascular disease in nursing practice: focus on children and youth: a scientific statement from the American Heart Association Committee on Atherosclerosis, Hypertension, and Obesity in Youth of the Council on Cardiovascular Disease in the Young, Council on Cardiovascular Nursing, Council on Epidemiology and Prevention, and Council on Nutrition, Physical Activity, and Metabolism. Circulation. 2007; 116:344–57. https://doi.org/10.1161/CIRCULATIONAHA.107.184595 PMID: 17592077
- Williams CL, Hayman LL, Daniels SR, Robinson TN, Steinberger J, Paridon S, et al. Cardiovascular health in childhood. Circulation. 2002; 106:143–60. <u>https://doi.org/10.1161/01.cir.0000019555.61092</u>. 9e PMID: 12093785
- Franks PW, Hanson RL, Knowler WC, Sievers ML, Bennett PH, Looker HC. Childhood obesity, other cardiovascular risk factors, and premature death. N Engl J Med. 2010; 362:485–93. https://doi.org/10. 1056/NEJMoa0904130 PMID: 20147714
- Berenson GS, Bogalusa Heart Study group. Health consequences of obesity. Pediatr Blood Cancer. 2012; 58:117–21. https://doi.org/10.1002/pbc.23373 PMID: 22076834
- Freedman DS, Khan LK, Serdula MK, Dietz WH, Srinivasan SR, Berenson GS. The relation of childhood BMI to adult adiposity: the Bogalusa Heart Study. Pediatrics. 2005; 115:22–7. <u>https://doi.org/10. 1542/peds.2004-0220 PMID: 15629977</u>
- 65. Ference BA, Ginsberg HN, Graham I, Ray KK, Packard CJ, Bruckert E, et al. Low-density lipoproteins cause atherosclerotic cardiovascular disease. 1. Evidence from genetic, epidemiologic, and clinical studies. A consensus statement from the European Atherosclerosis Society Consensus Panel. Eur Heart J. 2017; 38:2459–72. https://doi.org/10.1093/eurhearti/ehx144 PMID: 28444290
- Goldstein JL, Brown MS. A century of cholesterol and coronaries: from plaques to genes to statins. Cell. 2015; 161:161–72. https://doi.org/10.1016/j.cell.2015.01.036 PMID: 25815993
- Tilling K, Davies NM, Nicoli E, Ben-Shlomo Y, Kramer MS, Patel R, et al. Associations of growth trajectories in infancy and early childhood with later childhood outcomes. Am J Clin Nutr. 2011; 94:1808S– 13S. https://doi.org/10.3945/ajcn.110.001644 PMID: 21633072
- Howe LD, Tilling K, Benfield L, Logue J, Sattar N, Ness AR, et al. Changes in ponderal index and body mass index across childhood and their associations with fat mass and cardiovascular risk factors at age 15. PLoS ONE. 2010; 5(12):e15186. https://doi.org/10.1371/journal.pone.0015186 PMID: 21170348
- Higgins V, Asgari S, Chan MK, Adeli K. Pediatric reference intervals for calculated LDL cholesterol, non-HDL cholesterol, and remnant cholesterol in the healthy CALIPER cohort. Clin Chim Acta. 2018; 486:129–34. https://doi.org/10.1016/j.cca.2018.07.028 PMID: 30025755
- 70. Harder T, Bergmann R, Kallischnigg G, Plagemann A. Duration of breastfeeding and risk of overweight: a meta-analysis. Am J Epidemiol. 2005; 162:397–403. https://doi.org/10.1093/aje/kwi222 PMID: 16076830
- Gale C, Logan KM, Santhakumaran S, Parkinson JRC, Hyde MJ, Modi N. Effect of breastfeeding compared with formula feeding on infant body composition: a systematic review and meta-analysis. Am J Clin Nutr. 2012; 95:656–69. https://doi.org/10.3945/ajcn.111.027284 PMID: 22301930
- 72. Ramirez-Silva I, Rivera JA, Trejo-Valdivia B, Martorell R, Stein AD, Romieu I, et al. Breastfeeding status at age 3 months is associated with adiposity and cardiometabolic markers at age 4 years in Mexican children. J Nutr. 2015; 145:1295–302. https://doi.org/10.3945/jn.114.198366 PMID: 25926414
- Owen CG, Whincup PH, Odoki K, Gilg JA, Cook DG. Infant feeding and blood cholesterol: a study in adolescents and a systematic review. Pediatrics. 2002; 110:597–608. <u>https://doi.org/10.1542/peds.</u> 110.3.597 PMID: 12205266
- 74. Owen CG, Martin RM, Whincup PH, Smith GD, Cook DG. Does breastfeeding influence risk of type 2 diabetes in later life? A quantitative analysis of published evidence. Am J Clin Nutr. 2006; 84:1043–54. https://doi.org/10.1093/ajcn/84.5.1043 PMID: 17093156
- 75. Martin RM, Patel R, Kramer MS, Vilchuck K, Bogdanovich N, Sergeichick N, et al. Effects of promoting longer term and exclusive breastfeeding on cardiometabolic risk factors at age 11.5 years: a cluster-

randomized, controlled trial. Circulation. 2013; 129:321–9. https://doi.org/10.1161/CIRCULATIONAHA. 113.005160 PMID: 24300437

- 76. Kramer MS, Matush L, Vanilovich I, Platt RW, Bogdanovich N, Sevkovskaya Z, et al. Effects of prolonged and exclusive breastfeeding on child height, weight, adiposity, and blood pressure at age 6.5 y: evidence from a large randomized trial. Am J Clin Nutr. 2007; 86:1717–21. <u>https://doi.org/10.1093/ajcn/ 86.5.1717 PMID: 18065591</u>
- 77. Horta BL, Victora CG. Long-term effects of breastfeeding—a systematic review. Geneva: World Health Organization; 2013.
- Victora CG, Bahl R, Barros AJD, França GVA, Horton S, Krasevec J, et al. Breastfeeding in the 21st century: epidemiology, mechanisms, and lifelong effect. Lancet. 2016; 387:475–90. <u>https://doi.org/10.1016/S0140-6736(15)01024-7</u> PMID: 26869575
- Bhutta ZA, Das JK, Rizvi A, Gaffey MF, Walker N, Horton S, et al. Evidence-based interventions for improvement of maternal and child nutrition: what can be done and at what cost? Lancet. 2013; 382:452–77. https://doi.org/10.1016/S0140-6736(13)60996-4 PMID: 23746776